Author's response to reviews

Title: Evidence of what works to support and sustain care at home for people with dementia: a literature review

Authors:

Alison SF Dawson (a.s.f.dawson@stir.ac.uk)
Alison M Bowes (a.m.bowes@stir.ac.uk)
Fiona Kelly (f.kelly@bournemouth.ac.uk)
Kari D Velzke (k.d.velzke@stir.ac.uk)
Richard Ward (richard.ward1@stir.ac.uk)

Version: 3
Date: 22 January 2015

Author's response to reviews: see over
Dear Sirs/Madams,

‘Evidence of what works to support and sustain care at home for people with dementia: a literature review’ – revised manuscript

We are pleased to submit for your consideration a version of our manuscript which has been revised to take account of the comments of reviewer Frances Bunn, dated 7 January 2015, on the revised version of our original manuscript. In particular, we have:

1. Amended the text to remove any references to our work as a ‘systematic review’ in order to avoid any ambiguity
2. Removed a section of text regarding the detail of the literature review proforma as suggested

We look forward to hearing from you further as regards the manuscript in due course.

Regards,

Dr Alison Dawson

On behalf of the manuscript authors